Title: Case report: Acute Pancreatitis Caused By Postcholecystectomy Hemobilia.

Authors:

Halil Alis (halilalis@gmail.com)
Mehmet Abdussamet Bozkurt (msametbozkurt@yahoo.com)
Osman Zekai Oner Oner (osmanzekai@yahoo.com)
Kemal Dolay (kdolay@yahoo.com)
Ahmet Nuray Turhan (ahmetnturhan@gmail.com)
Adem Ucar (drademucar@yahoo.com)
Ercan Inci (drercaninci@beah.com)
Ersan Aygun (ersanaygun@yahoo.com)

Version: 4 Date: 3 April 2010

Author's response to reviews:

Firstfully we have corrected the gramer in our manuscript. We added the background, competing intrests, authors' contrubitions and concent sections to the manuscript and introduction, case presentation, discussion parts to the abstract. Units for the biochemical parameters were written. We added the deatils about ERCP and angiography according to the comments of the referees.CT figures were removed. We completed the 9th reference.Contact details that were absent for some of the authors were added.Also we are sending the written informed concent to you by email.